## BANCA D'ITALIA

# Temi di discussione

del Servizio Studi

### Prices, product differentiation and quality measurement: a comparison between hedonic and matched model methods

by Gian Maria Tomat



Number 547 - February 2005

The purpose of the Temi di discussione series is to promote the circulation of working papers prepared within the Bank of Italy or presented in Bank seminars by outside economists with the aim of stimulating comments and suggestions.

The views expressed in the articles are those of the authors and do not involve the responsibility of the Bank.

Editorial Board:

Giorgio Gobbi, Marcello Bofondi, Michele Caivano, Andrea Lamorgese, Francesco Paternò, Marcello Pericoli, Alessandro Secchi, Fabrizio Venditti, Stefania Zotteri, Raffaela Bisceglia (*Editorial Assistant*).

## PRICES, PRODUCT DIFFERENTIATION AND QUALITY MEASUREMENT: A COMPARISON BETWEEN HEDONIC AND MATCHED MODEL METHODS

by Gian Maria Tomat<sup>\*</sup>

### Abstract

The paper provides an analysis of the problems of construction of quality-adjusted price indexes within the framework of the theory of product differentiation. In the general case of price-making behaviour on the part of firms, hedonic regressions are defined on the basis of reduced forms of the equation relating equilibrium prices to product characteristics. The paper considers the reduced form given by the marginal cost function and shows that the Laspeyres hedonic price index provides a lower bound to the quality-adjusted rate of price change while the Paasche hedonic price index provides an upper bound to the quality-adjusted rate of price change. The properties of hedonic price indexes are compared with those of matched model indexes. The theory is applied to the study of personal computer prices in Italy during the 1995-2000 period.

JEL classification: C35, C43.

Keywords: discrete choice, price indexes, product quality, personal computer.

Contents	
1. Introduction	
2. The model	
3. Marginal costs and hedonic price indexes	
4. The data	
5. Estimation results	
6. Hedonic price indexes for PCs	
7. Matched model indexes	
8. Conclusions	
Tables and figures	
Appendix I	
References	

### Contents

<sup>\*</sup> Bank of Italy, Research Department, Rome Branch.

### **1. Introduction**<sup>1</sup>

The problem of how to account for quality change in the construction of measures of the rate of price change has been for a long time an important subject of research in economic measurement. The subject is of central importance for many fields of economic analysis, ranging from the study of movements of real economic activity to the problem of indexing nominal wages<sup>2</sup>.

The purpose of this paper is to provide an analysis of the problems of construction of quality-adjusted price indexes from the point of view of the theory of product differentiation. The original developments of the theory provided by Becker (1965) and Lancaster (1966) emphasized how it opened a new perspective on the treatment of new goods. Since the consumer choice problem was defined in the space of product characteristics, the introduction of new goods and the disappearance of old ones could be modeled in terms of changes in the consumption possibilities set, simplifying the problem of making welfare comparisons<sup>3</sup>.

Work in this field has subsequently benefited from several contributions from the industrial organization literature. Rosen (1974) provided an analysis of the product differentiation framework in a competitive equilibrium setting, modeling both consumer and firm choices as a function of product characteristics. Bresnahan (1981, 1987) extended the framework to allow for price-making behaviour on the part of firms and showed how the

<sup>&</sup>lt;sup>1</sup> Early versions of the paper where presented at the Bank of Italy-Istat workshop "Total Factor Productivity Measurement in Italy" held in January 2003 and at the Catholic University of the Sacred Heart of Milan in June 2003. Throughout the development of the work I received useful advice from Andrea Brandolini, Luigi Campiglio, Roberto Golinelli, Laura Leoni, Marco Magnani, Ariel Pakes, Giuseppe Parigi, Stefano Pisani, Stefano Siviero, Jack Triplett and Franceso Zollino. I would also like to thank Franceso Ferrari and Giorgio Panzeri for providing the price and characteristics data and Brian Gammage for providing the market shares data. The views contained in the paper are those of the author and do not necessarily reflect those of the Bank of Italy. Email address: gianmaria.tomat@bancaditalia.it.

<sup>&</sup>lt;sup>2</sup> Much of the recent interest in the problems of price measurement developed after the publication of the Boskin et. al. (1996) report, which analyzed the problems of measurement of the U.S. CPI. Recent reviews of the literature on price measurement include Lebow and Rudd (2003) and Wynne and Rodriguez-Palenzuela (2004). The latter work analyzes the problems of price measurement in the context of the European institutional framework.

model could be used to study the character of industry competition. More recently, Berry, Levinsohn and Pakes (1995, 2004) have provided further extensions of the basic model, drawing, among other subjects, on the field of discrete choice theory.

The implications of the theory of product differentiation for the construction of quality-adjusted price indexes, were analyzed initially by Griliches (1971), who used the theory to provide a rationale for the adoption of hedonic price indexes. Some of the implications of the more recent developments of the theory for the construction of hedonic price indexes are illustrated by Pakes  $(2003)^4$ .

The analysis of the properties of hedonic price indexes has so far been carried out on the basis of reduced forms of the hedonic function. The reduced form, however, has several problems of identification. In particular, in the reduced form demand and supply parameters are not identified and the unobservable components are generally characterized by crosssection dependence. These two features pose some problems of interpretation of estimates of the coefficients of hedonic regressions that should have implications for the possibility of using hedonic price indexes as measures of the quality-adjusted rate of price change.

In related contributions Pakes, Berry and Levinsohn (1993) and Nevo (2003) consider the problem of making exact welfare comparisons on the basis of the estimated parameters of consumer preferences in a discrete choice model. This approach has been used in these works to study problems such as the effect on consumer welfare of changes in environmental regulation or of the introduction of a new brand of a product. Specifying the consumer's choice problem in the space of product characteristics, simplifies the analysis in comparison with analogous problems studied on the grounds of traditional demand theory, as is done for example in works by Hausman (1997a, 1997b, 1999). However, the results of the above

<sup>&</sup>lt;sup>3</sup> Early versions of the theory of product differentiation were analyzed by Gorman (1956) and Dorfman, Samuelson and Solow (1958) in the form of the model known as the linear characteristics model.

<sup>&</sup>lt;sup>4</sup> Seminal works using hedonic price indexes to construct quality-adjusted price measures were provided during the 1960s by Griliches (1961), Triplett (1969) and Chow (1967). Interestingly, the former works were concerned with the measurement of prices of automobiles while the latter was concerned with computer prices and with the implications of price measurement for the study of the diffusion of computers in the U.S. economy. The literature that developed after these works is broad. Collection of studies developed for several sectors of the economy can be found in Jorgenson and Landau (1989), Gordon (1990) and Foss, Manser and Young (1993).

studies show that the approach can at most generate upper and lower bounds to the true price indexes and the empirical evidence shows that these bounds are usually quite wide.

The present paper follows a different line of research, it takes as point of departure the definition of price index provided by the hedonic regression approach and analyzes the properties of hedonic price indexes, compiled on the basis of estimates of the cost function of a partial equilibrium model. In this way the paper provides a partial resolution to the problems of identification that characterize hedonic regression models. The analysis shows that the Laspeyres hedonic price index provides on average a lower bound to the theoretical index while the Paasche hedonic price index provides an upper bound to the theoretical index.

The paper also illustrates how the proposed definitions of Laspeyres and Paasche price indexes can be used to study the behaviour of matched model indexes. It is shown that if matched products in reference and comparison periods are relatively close in characteristics space, the matched model indexes should not be characterized by serious distortions and should therefore be considered a useful alternative to the proposed hedonic indexes.

The analysis is applied to the study of personal computer prices in Italy during the 1995-2000 period. The paper is structured as follows. Section 2 reviews the theoretical framework. Section 3 uses the theory to define hedonic price indexes and analyzes their properties. Section 4 presents the data used for the analysis. Section 5 presents estimates of the partial equilibrium model used for the application with Italian personal computer data. Section 6 presents hedonic price indexes for personal computers in Italy in 1995-2000. Section 7 discusses the implications of the proposed framework for the compilation of matched model indexes and presents matched model indexes for personal computers in Italy for 1995-2000. Section 8 draws some conclusions.

### 2. The model

Following Berry (1994) and Berry, Levinsohn and Pakes (BLP) (1995, 2004) we consider a market for a differentiated commodity where there are M consumers and N varieties of the product. Different firms in the market produce different varieties of the product and each variety in the market is identified by a set of performance characteristics.

Consumers and firms observe all the product characteristics relevant for their choices, although some of the characteristics entering both consumer preferences and firm costs are unobserved by the econometrician. Let  $x_j \hat{I} R^K$  denote the observable characteristics of product *j*,  $x_j \hat{I} R$  the unobservable product characteristics influencing consumer choice,  $w_j \hat{I} R$  the unobservable characteristics entering the firm cost function and  $p_j \hat{I} R$  the price of product *j*, for  $j \hat{I} \{0, 1, ..., N\}$ , where j=0 denotes the outside alternative. For convenience, the price and characteristics of the outside alternative are normalized to zero and thus the market characteristics vectors and the market price vector are denoted as  $x=(x_1,...,x_N)\hat{I} R^{NK}$ ,  $x=(x_1,...,x_N)\hat{I} R^N$ ,  $w=(w_1,...,w_N)\hat{I} R^N$  and  $p=(p_1,...,p_N)\hat{I} R^N$ .

Consumers have heterogeneous preferences thus in each time period only a fraction of them will consume each product *j*. The market share function is denoted by  $s_j=s_j(x, \mathbf{x}, p, q)$ , for  $j\hat{\mathbf{I}}\{0, 1, ..., N\}$ , where q is a parameter representing consumer preferences. The market share of each product depends on the vectors of prices and characteristics of all products in the market and is derived in relation to the subset of the population of consumers for which consumption of product *j* is preference maximizing. The preference relation is defined in a way such that, for every vector  $(x, \mathbf{x}, p)\hat{\mathbf{I}}R^{N(K+2)}$  of prices and product characteristics, the market share of each product is greater than zero, market shares add up to unity and the market share function is continuously differentiable<sup>5</sup>.

Given that there are *M* consumers in the market, the demand for product *j* is given by:

(2.1) 
$$D_{i}(x, \mathbf{x}, p, \mathbf{q}) = Ms_{i}(x, \mathbf{x}, p, \mathbf{q})$$

Each product in the market is produced by a different firm. Since the model is used to describe the market for personal computers, which is a relatively young industry with many competing firms, this assumption does not appear to be restrictive in the present context. Marginal costs of production are a function of performance characteristics and the marginal

<sup>&</sup>lt;sup>5</sup> The distribution of preferences determining the market shares outcome is defined by a distribution of individual characteristics within the population of consumers. Formal derivations of the market share functions can be found in the literature for several specifications of the preference structure. See for example McFadden (1981) or Anderson, de Palma and Thisse (1992). In the more recent BLP (1995, 2004) framework, closed-form solutions of the market share functions cannot be derived and the analysis has to resort to more advanced methods, such as equilibrium analysis and simulation.

cost function is denoted as  $c(x_j, w_j, g)$ , where  $g\hat{I}R^K$  is a cost parameter. The marginal cost function is continuous in the characteristics vector  $(x_j, w_j)\hat{I}R^{(K+1)}$  and in the parameter  $g\hat{I}R^K$ .

In the general formulation of the model, the equilibrium is characterized by Bertrand pricing, so that firms set prices in order to maximize profits, taking prices of other firms as given. In equilibrium, price is equal to marginal cost plus a mark-up term, which depends on the price elasticity of demand:

(2.2) 
$$p_{j} = c(x_{j}, \boldsymbol{w}_{j}, \boldsymbol{g}) + \frac{D_{j}(x, \boldsymbol{x}, p, \boldsymbol{q})}{\left|\frac{\partial D_{j}(x, \boldsymbol{x}, p, \boldsymbol{q})}{\partial p_{j}}\right|}$$

The pricing equation forms the basis for the analysis of the hedonic price indexes. Since it relates price to product characteristics, it is equivalent to an hedonic regression function. However, in the general case of price-making behaviour on the part of firms, the pricing equation implies that in equilibrium the price of each product depends not only on its own characteristics but also on the price and characteristics of all the other products in the market. In prior research the hedonic regression analysis has been carried out mainly using reduced forms of the pricing equation, giving the price of each product as a function of its own performance characteristics. The next sections propose to use the cost function as a reduced form and analyze the properties of the price and quantity indexes resulting from this choice.

### 3. Marginal costs and hedonic price indexes

In order to define the hedonic price indexes consider two subsequent periods of time, a comparison period t-1 and a reference period t, and the question of how to compensate consumers for the price changes that occur between period t-1 and period t.

Consider initially defining the compensating variation for consumers of period t-1 product j as the difference between the price predicted for product j in period t on the grounds of period t cost conditions and its actual period t-1 price:

(3.1) 
$$CV_{i}(x_{it-1}, p_{it-1}, \boldsymbol{g}_{t}) = E(c(x_{it-1}, \boldsymbol{w}_{it-1}, \boldsymbol{g}_{t}) | x_{it-1}) - p_{it-1}$$

In (3.1)  $x_{jt-1}$  and  $p_{jt-1}$  are the observable characteristics and the market price of product j in period t-1,  $w_{jt-1}$  is the unobservable cost component in period t-1 and  $\gamma_t$  is the cost parameter in period t. The price predicted for period t-1 product j in period t is defined as the expected value of the marginal cost of production, given the observable characteristics of the product and the period t cost parameter. Given the set of expenditures prevailing in period t-1 and the set of prices prevailing in period t, equation (3.1) provides an estimate of the additional expenditure that consumers of product j should make in period t to consume the same product they were consuming in period t-1. Following Pakes (2003), note that since the set of goods changes between the two periods, if consumers are compensated for the price difference resulting from the changed market conditions, they may be able to make a different choice and improve their personal well-being. This definition of the compensation function can be applied properly only to period t-1 product t-1 product j there exists in period t at least one product with similar characteristics, so that (3.1) defines a meaningful compensation function.

An aggregate price index can be defined by averaging the compensating variations defined by equation (3.1) across the varieties of the differentiated commodity available in the comparison period that are within the reference period goods space. Let  $I_Z \mathbf{i} \{1,...,N\}$  index the relevant comparison period goods space and  $N_Z \mathbf{i}N$  denote the number of comparison period varieties in the reference period goods space. Giving an equal weight to each variety the Laspeyres hedonic price index is defined as:

(3.2) 
$$P_{HL}(x_{t-1}, p_{t-1}, \boldsymbol{g}_t) = \frac{1}{N_Z} \sum_{j \in I_Z} CV_j(x_{jt-1}, p_{jt-1}, \boldsymbol{g}_t)$$

Since in Bertrand pricing equilibrium the mark-up is in general greater than zero, it can be shown that equation (3.2) defines on average a lower bound to the true theoretical compensation function. Defining the compensating variation as a function of the observable cost component, one abstract from two other components of the true compensating price, an unobservable cost component and the mark-up term. The unobservable cost component is on average equal to zero and thus the mark-up term is what ultimately matters in defining the compensation error. Since by construction the mark-up is always greater than or equal to zero, so is the average compensation error.

*Proposition 3.1:* In the Bertrand pricing equilibrium defined by equation (2.2), the Laspeyres hedonic price index defined by equation (3.2) is a lower bound to the true price index for the average consumer of product j in period t-1.

Proof: mathematical appendix p. 33.

Next consider the problem from the perspective of period t consumers, in this case the compensation function is defined from the equivalent variation:

(3.3) 
$$EV_{j}(x_{jt}, p_{jt}, \boldsymbol{g}_{t-1}) = p_{jt} - E(c(x_{jt}, \boldsymbol{w}_{jt}, \boldsymbol{g}_{t-1} | x_{jt}))$$

where  $x_{jt}$  and  $p_{jt}$  are the observable characteristics and the market price of product j in period t,  $w_{jt}$  is the unobservable cost component in period t and  $g_{t-1}$  is the cost parameter in period t-1. The price predicted for period t product j in period t-1 is defined as the expected value of the marginal cost of production, given the observable characteristics of the product and the period t-1 cost parameter. Given the set of expenditures prevailing in period t and the set of prices prevailing in period t-1, equation (3.3) provides an estimate of the change in income that would allow period t consumers of product j to consume the same product in period t-1 and in period t. We note again that since the set of goods changes between the two periods of time, consumers are allowed to make choices that improve their personal well-being in period t-1. The equivalent variation can be applied properly only to period t products that are within period t-1 goods space, since it is based on the assumption that for each period t product j there exist in period t-1 at least one product with similar characteristics.

To define an aggregate price index, let as before  $I_Z$  index the relevant reference period goods space and  $N_Z$  denote the number of reference period varieties in the comparison period goods space. Averaging across varieties in the reference period that are within the comparison period goods space the Paasche price index is defined as:

(3.4) 
$$P_{HP}(x_t, p_t, \boldsymbol{g}_{t-1}) = \frac{1}{N_Z} \sum_{j \in I_Z} EV_j(x_{jt}, p_{jt}, \boldsymbol{g}_{t-1})$$

Since in Bertrand pricing equilibrium the mark-up is in general greater than zero, equation (3.4) defines on average an upper bound to the true theoretical compensation function.

*Proposition 3.2:* In the Bertrand pricing equilibrium defined by equation (2.2), the Paasche hedonic price index defined by equation (3.4) is an upper bound to the true price index for the average consumer of product j in period t.

### *Proof:* mathematical appendix p. 35.

Finally, note that despite the bounding properties provided by propositions 3.1 and 3.2 there is in general no relation between the Laspeyres and Paasche hedonic indexes since they are defined with reference to two different sets of goods and two different sets of consumers, the period t-1 sets for the Laspeyres index and the period t sets for the Paasche index. However, in the limiting case of an unchanged set of goods between the two periods and if consumer preferences do not change over time it can be easily shown that  $P_{HP}(x_{t},p_{t},\mathbf{g}_{-1})^{3}P_{HL}(x_{t-1},p_{t-1},\mathbf{g})$ . Provided the sets of goods for the two indexes are not too different and if variations in preferences can be neglected this relation should hold more generally.

### 4. The data

Following previous research on computer prices by Aizcorbe, Corrado and Doms (2000), Berndt and Griliches (1993), Berndt, Griliches and Rappaport (1995), Berndt and Rappaport (2001) and Moch (2001) data on prices and characteristics of personal computers traded in Italy during the 1995-2000 period were collected from the monthly issues of the Italian editions of PC Professional and PC Magazine.

Table 1 reports sample averages for price and for some of the main performance characteristics of the PCs included in our sample. The table shows that while prices remained roughly constant over the sample period, the distribution of the main desktop characteristics shifted markedly to the right. This is true for both continuous characteristics such as frequency, RAM, hard disk capacity, video memory and monitor size and for indicator variables of computer accessories such as CD player, DVD player, modem and

audio interface. Moreover, the quality of these devices also increased substantially during the sample period, as is shown by the increase in their measures of speed.

Table 2 presents for each year the distribution of personal computers by processor type. In the first year of the sample period more than half of all computers were already equipped with a Pentium processor while the remaining ones were still using the 486DX2 and 486DX4 processors. In the subsequent years, the latter types of processors gradually disappeared from the market, while the Pentium technology was gaining market shares. Towards the end of the sample period, the share of personal computers equipped with a Pentium processor declined to less than 50 per cent, while the market share of new processor types such as Athlon and Celeron was increasing.

Table 3 reports for each year the distribution of computers by computer brand. The main computer brands we identify in our sample are Acer, Compaq, Dell, Digital Equipment, Hewlett Packard, Ibm and Olivetti. Throughout the sample period, the cumulative share of the main computer brands is of the order of 5 per cent. Most of the personal computers in our sample are thus produced by small to medium sized specialized suppliers, reflecting the structure of the Italian computer industry.

Market shares for each personal computer in the price and characteristics database, were estimated using information on market shares by processor type provided by the Gartner Group corporation. Letting N denote the number of computers in a given year in our sample,  $\overline{w}_i$  the sample share of processor i as reported in Table 2,  $w_i$  the market share of processor i provided by Gartner Group, the market share of each computer in our sample equipped with processor *i* was estimated as  $(1/N)(w_i/\overline{w}_i)$ . The factor  $(w_i/\overline{w}_i)$  corrects for over or under sampling of computers with a particular processor type. The estimated shares of each computer *j* were then multiplied by the share of personal consumption expenditure in computer processing equipment to get an estimate of the shares  $s_i$ .

### 5. Estimation results

For estimation we assume that demand takes the form of the nested logit model of McFadden (1978, 1984) and Cardell (1997). This model is based on the assumption that

there are G product groups and that preferences for products belonging to each group are characterized by a common random component, which varies between individuals and groups, in addition to the random component of the simple logit model, which varies between individuals and products. Denoting with  $s_{j/g}(x, \mathbf{x}, p, \mathbf{q})$  the within-group market share functions, for  $j\hat{\mathbf{I}}g$  and  $g\hat{\mathbf{I}}\{1,...,G\}$ , and provided preferences are linear in prices and product characteristics, the market share functions satisfy the following relation:

(5.1) 
$$\ln s_j - \ln s_0 = \mathbf{b} x_j - \mathbf{a} p_j + \mathbf{s} \ln s_{j/g} + \mathbf{x}_j$$

where a>0 is a parameter determining the own-price elasticity of demand and  $0 \pounds s < 1$  is a parameter determining the relative importance of the common and idiosyncratic random components of consumer preferences and shaping the cross-price elasticities.

We also assume that marginal costs are a linear function of performance characteristics:

(5.2) 
$$c(x_j, \boldsymbol{w}_j, \boldsymbol{g}) = \boldsymbol{g} x_j + \boldsymbol{w}_j$$

These assumptions imply that the pricing equation takes the following form:

(5.3) 
$$p_j = \mathbf{g} \mathbf{x}_j + \left(\frac{1-\mathbf{s}}{\mathbf{a}} \frac{1}{1-\mathbf{s} \mathbf{s}_{j/g} - (1-\mathbf{s}) \mathbf{s}_j}\right) + \mathbf{w}_j$$

For s=0 the model reduces to the standard logit model of McFadden (1974, 1984), where the common random component is absent, for s @ 1 the weight of the idiosyncratic component tends to zero and the correlation between the within-group innovations tends to one<sup>6</sup>.

In order to define the partition of the product space, for each year of the 1995-2000 sample period products are ordered according to their price and the product groups are defined by the quartiles of the price distribution. This definition of the partition can be

<sup>&</sup>lt;sup>6</sup> The nested logit model allows for a less restrictive pattern of elasticity than the simple logit model, although it requires the definition of a set of product groups. The more recent BLP (1995, 2004) models provide more general formulations of the consumer preference structure, which avoid the requirement of the definition of a partition of the product space.

justified assuming that consumers with different incomes have different preferences regarding the allocation of expenditure between personal computers and the outside good.

Given the information set, equations (5.1) and (5.3) together with some additional distributional assumptions can be used for estimation. We follow the more recent literature and estimate the model using the generalized method of moments (GMM). This method requires the additional assumptions of orthogonality between the unobservable and the observable characteristics variables. Letting  $\mathbf{u}_i = (\mathbf{x}_i, \mathbf{w}_i)$  we thus assume:

(5.4) 
$$E(\mathbf{u}_i | x) = 0 \quad j \in \{1, ..., N\}$$

Since the moment conditions are non-linear in the a and s parameters, estimation cannot be performed using traditional instrumental variable methods. Moreover, the estimation algorithm cannot be based on traditional gradient methods, such as Newton-Raphson, since these methods cannot in general determine the global optimum of a function characterized by more than one local optima. Preliminary experimentation with the nested logit model using gradient methods shows that, for given different initial conditions within the relevant parameter range, the GMM estimator converges to different optima. We therefore adopt a "concentrated" GMM procedure. The estimation algorithm is based on stochastic search and is articulated in two stages. In the first stage, for each value of the nonlinear parameters a and s, the linear parameters are concentrated out of the criterion function using Hansen's (1982) two-step heteroskedasticity consistent estimator. In the second stage, the resulting concentrated criterion is optimized on the parameter space for aand s, using the method of simulated annealing described in Goffe, Ferrier and Rogers (1994)<sup>7</sup>.

In model estimation we could not use all the available characteristics variables. Given the number of observations for each year and the character of the estimation procedure the model in this case would be overparametrized. We therefore resort to a specification including continuous variables for frequency, RAM, hard disk capacity and video memory

<sup>&</sup>lt;sup>7</sup> Simulated annealing has been found to be superior to other global optimization methods, such as simplex search, for the optimization of functions characterized by multiple optima. Although on conventional

and indicator variables for CD player, DVD player, modem device and audio interface. Moreover, given the limited number of observations available for each year, in order to reduce finite sample distortion we do not use all the available moment conditions. We include in the set of instruments only the characteristics variables used in the specification of the market share and pricing equations and indicator variables for processor type<sup>8</sup>.

Table 4 reports some of the main results of estimation for two empirical specifications of the model. In the base specification, for each observation only the observable characteristic variables of the same observation are used as instruments. In the augmented specification, the characteristics variables of the preceding and subsequent observations in the rank ordering by product price are also used as instruments. In this way we control for the behaviour of the efficiency of the estimator as the number of moment conditions increases. The table shows that in the base specification there is a small number of overidentifying restrictions and the Sargan test rejects the model at a 5 per cent significance level in almost all years. In the augmented specification instead, the number of overidentifying restrictions is more in line with the number of moment conditions increases the adjusted  $R^2$  of both the market share and the pricing equations typically improves, although by a small amount<sup>9</sup>.

Table 5 reports estimates of the pricing equation implied by the parameter estimates for the augmented specification of the model. For each year, the price and estimates of the mark-up, the marginal cost and the mark-up rates are provided for the personal computers corresponding to the first quartile, the mean and the third quartile of the price distribution.

problems the algorithm is relatively slow, it is characterized by relatively good convergence properties. A formal description of the algorithm can be found in Judd (1998).

<sup>&</sup>lt;sup>8</sup> For computational convenience we do not make any attempt to look for optimal instruments. The criteria used to choose the instruments is that of maximizing their correlation with the independent variables in order to minimize the estimation variance.

<sup>&</sup>lt;sup>9</sup> For ease of exposition we do not report the estimates of the parameters of the market share and pricing equations. We note that most of the coefficients of the characteristics variables turned out significant and with correct signs in both equations in all years. Moreover, the estimated parameters for the characteristics variables remain relatively stable in going from the base to the augmented specification, while their standard errors do not show any sensible improvement. The improvement in efficiency that is implied by the adjusted  $R^2$  comparisons seems thus to be due entirely to a better identification of the non-linear parameters and therefore of the mark-up and the demand elasticity.

The price levels and the mark-up rates are reproduced graphically in Figures 1 and 2. The table and Figure 1 show that there was an increase in computer prices following the exchange rate shocks that occurred in 1995. Prices increased sharply in 1996 and then returned to normal levels in the following years. The table and Figure 2 also show that the mark-up is estimated to be close to zero at the beginning of the sample period, to increase substantially in 1997 when mark-up rates rise to more than 20 per cent and to decrease thereafter, returning close to zero at the end of the sample period.

The reported results provide interesting information on the behaviour of the Italian personal computer industry during the 1995-2000 period. In 1995 and 1996 it is likely that the devaluation of the Italian currency, which occurred as Italy was rejoining the European exchange rate mechanism, caused a contraction of mark-up rates for Italian personal computer producers while the impact of higher computer component costs was partly translated into higher prices and lower competition. Similarly, the developments of international financial markets that led to the devaluations of the East Asian currencies in the autumn of 1997 seem to have acted as a favourable cost shock to the personal computers industry in Italy, fostering competition and inducing firms to increase their mark-ups. The model estimates provide evidence of significant market dynamics for 1997 and subsequent years<sup>10</sup>.

### 6. Hedonic price indexes for PCs

With the background of the data and the analysis presented in the previous sections, we compiled hedonic price indexes for personal computers in Italy for the 1995-2000 period. The first panel of Table 6 reports the rates of change of both Laspeyres and Paasche price indexes, compiled as defined in section 3 of the paper. For each type of index, the table presents both the weighted and the unweighted versions. In the weighted forms, each product is weighted according to its market share instead of receiving an equal weight in the index.

<sup>&</sup>lt;sup>10</sup> This interpretation is supported by evidence on the behaviour of aggregate prices and mark-ups of the Italian economy over a relatively long period covering the 1980s and the early 1990s. As illustrated by Visco (1994) on the basis of estimates of the Bank of Italy, macroeconomic quarterly model, Italian producers tend to reduce mark-ups and increase prices in response to unfavourable exchange rate shocks and to increase mark-ups in response to favourable exchange rate shocks.

In weighted Laspeyres indexes, period t-1 product shares are used as weights and in weighted Paasche indexes, period t market shares are used as weights. In this way we assess the robustness of the results to the sampling variation resulting from the choice of sources for the personal computers data on prices and characteristics. Weighted indexes correct for the finite sample distortion resulting from sampling variation but the asymptotic properties of the indexes are not affected by this choice. The reported indexes are proper, for the Laspeyres price indexes, for each comparison period only products within the sample space defined by the reference period continuous variables used in model estimation are included in the computations and for the Paasche price indexes, for each reference period only products within the goods space defined by the comparison period continuous variables are considered. The match rate for the Laspeyres indexes is relatively good. In most years more than 50 per cent of the products can be included in the computations. For the Paasche price indexes the match rate is generally lower<sup>11</sup>.

The hedonic price indexes show that during the sample period there have been large decreases in PC prices on a quality-adjusted basis. The average yearly rate of price change over the sampling period is equal to -0.354 for the unweighted Laspeyres price index and to -0.355 for the weighted index. The differences in the yearly rates of change of the weighted and unweighted indexes appear to be small. The Paasche price indexes show higher rates of price change than the Laspeyres indexes, the average yearly rate of price change for the unweighted index is -0.119 while the average rate of change for the weighted index is -0.131. For the Paasche price indexes in some years the distortion of the unweighted index appears to be larger. Nevertheless the rates of change of the Paasche indexes are higher than the rates of change of the corresponding Laspeyres indexes in all years.

The second panel of Table 6 reports adjusted hedonic price indexes. The compensating and equivalent variations for these indexes are computed by adding to the estimated observed cost components the estimated unobserved cost components. For the Laspeyres price indexes the unobserved cost components refer to period t-1 and for the Paasche price

<sup>&</sup>lt;sup>11</sup> We do not impose sample space constraints for indicator variables since in some cases these may be too binding. For example, consumers may substitute a CD player with a DVD player or a modem with an audio interface. Requiring perfect matches on these devices would therefore substantially reduce the match rate, introducing a sample selection distortion in the price indexes.

indexes to period t. In this way the adjusted indexes remove the finite sample distortion arising from the unobservable cost components. Adjusted hedonic indexes tend to show lower rates of price change than hedonic ones, the average annual rate of price change is -0.370 for the unweighted adjusted hedonic Laspeyres index and -0.369 for the weighted index. The yearly rates of change are lower than the rates of change of the corresponding hedonic indexes in almost all years. The effect of weighting is relatively small. The average rate of price change is equal to -0.215 for the unweighted Paasche index and to -0.197 for the weighted Paasche index and the yearly rates of change are lower than the corresponding hedonic Paasche indexes in almost all years. The rates of change of the adjusted Paasche indexes are higher than the rates of change of the adjusted Paasche indexes in all years.

### 7. Matched model indexes

In order to provide an assessment of the results of the hedonic regression analysis, in this section we use the price and characteristics data to compile matched model indexes. To provide a definition of a matched model index, we note that since the set of available goods changes over time and in each period the characteristics of the available goods include many observable and unobservable components, we can proceed by assuming that the set of goods available in period *t* is completely different from the set available in period t-1.

Considering the problem of measurement from the point of view of period t-1 consumers, provided we restrict the analysis to the set of period t-1 products that are within the period t goods space, we could compile a price index by matching each period t-1 product with the period t product with the closest characteristics. The Laspeyres matched model index could thus be defined as:

(7.1) 
$$P_{ML}(\hat{p}_t, p_{t-1}) = \frac{1}{N_Z} \sum_{j \in I_Z} (\hat{p}_{jt} - p_{jt-1})$$

where  $\hat{p}_{jt}$  is the price of the product in period t goods space that is closest in observable characteristics to the period t-1 product j and  $\hat{p}_t = (\hat{p}_{1t}, ..., \hat{p}_{Nt})$ .

Similarly, for period t consumers and restricting the analysis to the set of period t products that are within the period t-1 goods space, we could compile a price index by

matching each period t product with the period t-1 product with the closest characteristics. The Paasche matched model index could then be defined as:

(7.2) 
$$P_{MP}(p_t, \hat{p}_{t-1}) = \frac{1}{N_Z} \sum_{j \in I_Z} \left( p_{jt} - \hat{p}_{jt-1} \right)$$

where  $\hat{p}_{jt-1}$  is the price of the product in period *t*-1 goods space that is closest in observable characteristics to the period *t* product *j* and  $\hat{p}_{t-1} = (\hat{p}_{1t-1}, ..., \hat{p}_{Nt-1})$ .

Matched model indexes are characterized by distortions of a different nature from the ones characterizing hedonic price indexes, the distortions depend on the rate of technological change. In general, products in the comparison period tend to be matched with reference period products with higher characteristics and products in the reference period tend to be matched with products of lower characteristics. Therefore both Laspeyres and Paasche indexes have a tendency to show an upward bias, which is an increasing function of the rate of product innovation. However, note that since matched model indexes are compiled by sampling prices directly rather than by estimating them as in hedonic indexes, they are not characterized by the distortions characterizing the latter ones. Provided a good quality of the matches between one period and the next is ensured, they should therefore provide a useful alternative to hedonic price indexes.

The matched model indexes are reported in the third panel of Table 6, both the Laspeyres and the Paasche indexes are compiled with reference to the same sample spaces used in the compilation of the hedonic price indexes. The panel shows that over the 1995-2000 period, both the weighted and the unweighted Laspeyres matched model indexes decline at an annual average rate of -0.252, the Paasche matched model index declines at an average rate of -0.122 and the Paasche matched model weighted index declines at an average annual rate of -0.145. Similarly to the hedonic indexes, the Paasche indexes are more sensitive to the choice of weights than the Laspeyres indexes. Comparing the matched model indexes with the hedonic price indexes, we note that the rates of change of the Laspeyres indexes in almost all years. The rates of change of the Paasche price index instead tend to be lower than the rates of change of the corresponding hedonic index and higher than

the rates of change of the corresponding adjusted hedonic indexes. This holds for both weighted and unweighted indexes.

It is interesting to compare these indexes with the official index for information processing equipment produced by Istat for the compilation of the CPI. The official index is compiled using a matched model method and during the 1995-2000 period declines at an average rate of -0.077. This is much lower than the average rates of the indexes compiled in the present paper. The distortion partly reflects a distortion of the index for the years 1995-98. During this period the index was compiled using a fixed base approach and declined at an average rate of -0.010. In 1998 Istat adopted a chain index system for the compilation of the CPI and the index displays a rate of price decline equal to -0.144 in 1998-99 and to -0.195 in 1999-2000. The bias of the official index is thus mainly the result of a failure of the fixed base approach to correct properly for quality change.

Summarizing the results obtained so far, note that consistently with the predictions of the theory, the Paasche hedonic indexes tend to show higher rates of price change than the Laspeyres hedonic indexes. This is true for both the weighted and the unweighted versions. The same result holds for adjusted hedonic indexes. Moreover, the Laspeyres matched model indexes show higher rates of change than the corresponding hedonic indexes while the Paasche matched model indexes show lower rates of change than the hedonic Paasche indexes and higher rates of change than the adjusted hedonic price indexes. These results are also broadly in line with the theory. Note that both hedonic and matched model Paasche indexes seem to display a poorer performance compared with the corresponding Laspeyres indexes. In particular, the finite sample corrections for product weighting and for the presence of unobservables seem to have a greater influence on the behaviour of the indexes. This result could be explained by the lower match rate that is observed for the Paasche indexes since the effect of introducing finite sample corrections has a tendency to be stronger for years where the match rate is lower. A low match rate implies that the indexes are characterized by greater sample selection and by greater uncertainty, making them less interpretable. Finally, looking closely at the data, we find that for both Laspeyres and Paasche matched model indexes product matchings are usually made between goods that are very close in characteristics space. This implies that the indexes should not be characterized by a substantial innovation bias.

### 8. Conclusions

The previous sections provide an analysis of the problems of construction of qualityadjusted price indexes using the theory of product differentiation. The analysis focuses on the recent developments of the theory, which extend prior attempts to model a partial equilibrium with product differentiation.

The general model defines consumer and firm choices in the space of product characteristics and allows for price-making behaviour on the part of firms. The industry equilibrium is stated in terms of a system of simultaneous equations, giving the market share and the equilibrium price of each product as a function of performance characteristics. The pricing equation of the partial equilibrium model forms the basis for the hedonic regression analysis. In the general case of Bertrand pricing, the equilibrium price of each product depends both on its own performance characteristics and on the prices and performance characteristics of all other products in the market.

The approach followed most often in the quality change literature has been to define an hedonic regression function using reduced forms of the pricing equation. The paper proposes to use as a reduced form the marginal cost function of the partial equilibrium model and analyzes the properties of the resulting hedonic price indexes. The Laspeyres hedonic price index is shown to provide a lower bound to the true rate of price change and the Paasche price index is shown to provide an upper bound to the true rate of price change. The paper also provides definitions for Laspeyres and Paasche matched model indexes. These are given according to the theoretical concepts for the quality-adjusted rate of price change implied by the model. Since the set of products available in the market usually improves in quality over time, matched model indexes tend to be characterized by an upward bias. However, provided matches are made between products that are close in characteristics space, the matched model indexes should provide a useful alternative measure of the rate of price change.

The paper contains an application of the theory to the study of personal computer prices in Italy during the period 1995-2000. The application is developed on the basis of data on prices, characteristics and market shares of personal computers available in the Italian market during the sample period, collected from personal computer magazines and private research institutions. The data show that the computer market was very dynamic during the

period, personal computers experienced substantial increases in performance characteristics while prices remained roughly constant.

The data are used to estimate a specification of the partial equilibrium model based on nested logit preferences. Following the more recent research, estimation is based on the generalized method of moments. The non-linear character of the model leads to the adoption of a concentrated version of the generalized method of moments procedure, based on the global optimization algorithm of simulated annealing. Estimation leads to satisfactory results, the coefficients of the share and pricing equations are usually significant and have the correct sign and the estimates of the pricing equation reflect in an interpretable way the history of the shocks affecting the Italian personal computer industry during the period.

The estimated nested logit model and the data are then used to compile both Laspeyres and Paasche hedonic price indexes. The hedonic price indexes show that the rate of price decline for personal computers in Italy during the sample period was considerable. The results are consistent with findings for other countries and the indexes seem to conform to the properties implied by the theory. The paper also presents Laspeyres and Paasche matched model indexes compiled with the same data used in the hedonic regression analysis. Conforming to prior expectations, the Laspeyres matched model indexes show rates of price decline that are usually lower than the corresponding hedonic indexes and the Paasche matched model indexes show rates of price decline that are higher than the corresponding hedonic indexes, though for the Paasche indexes the results are less clear cut due to the lower match rate which is observed for these indexes. Moreover, the matched model indexes do not seem to be characterized by any substantial innovation bias.

The empirical results suggest that hedonic and matched model methods should in general be thought of as complementary alternatives. While hedonic price indexes are characterized by distortions depending on the properties of the reduced forms of the hedonic function, matched model indexes are characterized by the classical distortions arising from the requirement of making appropriate sample replacements. There is no prior reason why one method should perform better than the other, rather the theory suggests that the choice between the two methods is essentially an empirical matter.

Table 1

	1995	1996	1997	1998	1999	2000
Price	4'203	5 <b>'</b> 242	4'223	4 <b>'</b> 165	4'130	4′319
Frequency	89	148	185	296	468	740
RAM	9	21	27	57	104	121
Hard disk capacity	0.6	1.7	2.5	5.2	9.8	21.3
CD	0.383	0.824	1.000	0.928	0.495	0.298
CD*speed	1	5	14	26	31	38
DVD	0.000	0.000	0.000	0.027	0.505	0.690
DVD*speed	0	0	0	1	3	8
Video memory	2	2	3	5	17	32
Modem	0.000	0.066	0.364	0.288	0.729	0.536
Modem*speed	0.0	1.8	12.8	15.3	40.9	30.0
Monitor size	15.3	15.7	15.3	15.9	16.6	16.7
Audio	0.298	0.549	0.798	0.910	0.981	0.952
Observations	47	91	99	111	107	84

### SAMPLE AVERAGES FOR PRICE AND CHARACTERISTICS $\mbox{DATA}^{(1)}$

(1) Prices inclusive of the monitor price and of value added tax in thousands of lira, frequency in Mhz, RAM in Mbytes, hard disk capacity in Gbytes, video memory in Mbytes, monitor size in inches. The variables CD, DVD, modem and audio are indicator variables. CD and DVD speed in standard units of reading speed, each unit of reading speed is equal to 150 Kbytes per second. Modem speed in Kbytes per second.

	1995	1996	1997	1998	1999	2000
486DX2	0.191	0.000	0.000	0.000	0.000	0.000
486DX4	0.213	0.011	0.000	0.000	0.000	0.000
Athlon	0.000	0.000	0.000	0.000	0.019	0.369
Celeron	0.000	0.000	0.000	0.117	0.140	0.119
Duron	0.000	0.000	0.000	0.000	0.000	0.048
K6-3	0.000	0.000	0.000	0.000	0.019	0.000
Pentium	0.596	0.978	0.283	0.000	0.000	0.000
Pentium II	0.000	0.000	0.030	0.820	0.262	0.000
Pentium III	0.000	0.000	0.000	0.000	0.561	0.464
Pentium MMX	0.000	0.000	0.687	0.063	0.000	0.000
Pentium Pro	0.000	0.011	0.000	0.000	0.000	0.000
Observations	47	91	99	111	107	84

DISTRIBUTION OF PERSONAL COMPUTERS BY PROCESSOR TYPE

Table 3

### DISTRIBUTION OF PERSONAL COMPUTERS BY BRAND<sup>(1)</sup>

	1995	1996	1997	1998	1999	2000
	0.000	0.011	0.010	0.000	0.000	0.024
Acer	0.000	0.011	0.010	0.000	0.000	0.024
Compaq	0.000	0.000	0.000	0.010	0.010	0.024
Dell	0.000	0.000	0.000	0.010	0.010	0.012
Digital Equipment	0.000	0.011	0.010	0.000	0.000	0.000
Hewlett Packard	0.000	0.011	0.030	0.000	0.000	0.012
Ibm	0.021	0.022	0.020	0.010	0.000	0.000
Olivetti	0.000	0.011	0.010	0.000	0.010	0.000
Observations	47	91	99	111	107	84

(1) Main personal computer brands.

	1995	1996	1997	1998	1999	2000
			Base spe	cification		
Demand adj. R <sup>2</sup>	0.748	0.024	0.172	0.035	0.107	0.017
Supply adj. R <sup>2</sup>	0.267	0.204	0.199	0.258	0.661	0.640
Degrees of Freedom	5	6	9	9	14	11
Sargan test (P-value)	0.104	0.026	0.009	0.000	0.003	0.047
Observations	46	89	95	110	107	67
	Augmented specification					
Demand adj. R <sup>2</sup>	0.130	0.012	0.177	0.044	0.109	0.149
Supply adj. R <sup>2</sup>	0.290	0.272	0.180	0.322	0.662	0.733
Degrees of Freedom	23	45	46	57	65	34
Sargan test (P-value)	0.123	0.284	0.401	0.322	0.103	0.315
Observations	44	83	87	106	105	54

### SUMMARY STATISTICS FOR THE NESTED LOGIT MODEL<sup>(1)</sup>

(1) Mean utility levels and marginal cost functions are linear in the performance characteristics. Characteristics variables are frequency, RAM, hard disk capacity, video memory, CD, DVD, modem and audio. Instrumental variables include indicator variables for processor type. Base specification for each observation uses as instruments only variables of the same observations. Augmented specification uses also variables of the preceding and succeeding observations in the ordering by product price.

	1995	1996	1997	1998	1999	2000		
			$1^{st}$	qrt.				
Price	3′749	4'101	3′510	3'180	3′030	2′951		
Mark-up	0	0	784	295	28	0		
Marginal cost	3′748	<b>4'</b> 101	2′726	2′885	3'002	2'950		
Mark-up rate	0.000	0.000	0.287	0.102	0.009	0.000		
		Mean						
Price	4′203	5 <b>′</b> 242	4′223	4′165	4′130	4′319		
Mark-up	0	0	784	295	28	0		
Marginal cost	4'203	5′242	3′439	3′870	4'102	4′319		
Mark-up rate	0.000	0.000	0.228	0.076	0.007	0.000		
	3 <sup>rd</sup> qrt.							
Price	4′582	6′057	4′690	5′058	5′508	5′148		
Mark-up	0	0	784	295	28	0		
Marginal cost	4′581	6′057	3′906	4′763	5′480	5'148		
Mark-up rate	0.000	0.000	0.201	0.062	0.005	0.000		

PRICING EQUATIONS AND MARK-UP RATES<sup>(1)</sup>

(1) Prices, mark-ups and marginal costs in thousands of lira. Personal computers at the first quartile, the mean and the third quartile of the price distribution.



### **PRICES OF PERSONAL COMPUTERS**<sup>(1)</sup>

(1) Prices in thousands of lira. Personal computers at the first quartile, the mean and the third quartile of the price distribution.

Figure 2



### MARK-UP RATES<sup>(1)</sup>

(1) Personal computers at the first quartile, the mean and the third quartile of the price distribution.

Table 6

### PRICE INDEXES FOR PERSONAL COMPUTERS<sup>(1)</sup>

	1995-96	1996-97	1997-98	1998-99	1999-2000	Average 1995-2000	
	Hedonic indexes						
Laspeyres Weighted Laspeyres	-0.219 -0.216	-0.416 -0.432	-0.239 -0.239	-0.498 -0.492	-0.355 -0.351	-0.354 -0.355	
Observations	35	53	15	48	59		
Paasche Weighted Paasche	-0.057 -0.057	-0.279 -0.258	0.194 -0.055	-0.265 -0.265	-0.110 0.021	-0.119 -0.131	
Observations	27	72	43	27	7		
		1	Adjusted h	edonic ind	exes		
Laspeyres Weighted Laspeyres	-0.237 -0.235	-0.436 -0.436	-0.180 -0.180	-0.522 -0.522	-0.409 -0.407	-0.370 -0.369	
Observations	34	49	14	46	58		
Paasche Weighted Paasche	-0.159 -0.159	-0.297 -0.268	0.102 -0.003	-0.286 -0.287	-0.360 -0.235	-0.215 -0.197	
Observations	25	66	41	26	4		
	Matched model indexes						
Laspeyres Weighted Laspeyres	-0.163 -0.159	-0.254 -0.273	-0.049 -0.049	-0.518 -0.512	-0.184 -0.171	-0.252 -0.252	
Observations	35	53	15	48	59		
Paasche Weighted Paasche	0.069 0.069	-0.161 -0.248	-0.094 -0.240	-0.431 -0.422	0.130 0.291	-0.122 -0.145	
Observations	27	72	43	27	7		

(1) Annual rates of change, compiled using the ratio of arithmetic means formula. Laspeyres indexes use comparison period products, Paasche indexes use reference period products. Weighted indexes use market shares to weight each product in the sample. Adjusted hedonic indexes remove the finite sample distortion arising from the unobservable cost components.

### Appendix I

### **Mathematical Appendix**

We provide here a formal derivation of the propositions made in the text. In order to establish notation and some preliminary results let  $X_s = (X_{1s}, ..., X_{Ns}) \hat{I} R^{NK}$ ,  $X_s = (X_{1s}, ..., X_{Ns}) \hat{I} R^N$ ,  $W_s = (W_{1s}, ..., W_{Ns}) \hat{I} R^N$  and  $P_s = (P_{1s}, ..., P_{Ns}) \hat{I} R^N$  denote respectively the random vector corresponding to the sets of observable characteristics, unobservable preference characteristics, unobservable cost characteristics and to the set of prices of the goods available in period *s*. The observables are *K*-dimensional so that  $X_{js} \hat{I} R^K$  for all  $j \hat{I} \{1, ..., N\}$  and  $s \hat{I} \{t-1, t\}$ . We follow the convention of denoting realizations of random vectors by lower case letters.

Laspeyres and Paasche price indexes have been defined in the main text with reference to products belonging to subsets of the relevant product space. To study the properties of the indexes it is therefore necessary to establish the sampling properties of products drawn in subsets of the product space.

Let  $F_{js}(x_{js})$  denote the distribution function of the random vector  $X_{js}$ , for  $j\hat{I}\{1,...,N\}$  and  $s\hat{I}\{t-1,t\}$ , and for a subset  $Z_j\hat{I}R^K$  of the characteristics space let  $P(x_{js} \in Z_j) = \int_{Z_j} dF_{js}(x_{js})$  denote the probability that  $x_{js}$  belongs to  $Z_j$ . Similarly, let  $F_s(x_s)$  be the distribution function of the random vector  $X_s$ , for  $s\hat{I}\{t-1,t\}$ , and for a subset  $Z\hat{I}R^{NK}$  of the product space let  $P(x_s \in Z) = \int_Z dF_s(x_s)$  denote the probability that  $x_s$  belongs to Z. The probability and distribution functions for the random variables  $X_{js}$  and  $W_{js}$  are defined in a similar way.

We make the following assumption about the distribution of the random variables  $X_{js}$ ,  $X_{js}$  and  $W_{js}$ .

Assumption I.1: In each period of time  $s\hat{I}\{t-1,t\}$ , the random variables  $X_{js}$ ,  $X_{js}$  and  $W_{js}$  are identically and independently distributed for all  $j\hat{I}\{1,...,N\}$ .

From assumption I.1 it follows that  $F_{js}(x_{js}) = F_{ks}(x_{ks})$  for all  $j,k\hat{I}\{1,...,N\}$ ,  $s\hat{I}\{t-1,t\}$  and  $x_{js}=x_{ks}$  and  $F_s(x_s) = \prod_{i=1}^N F_{js}(x_{js})$  for  $s\hat{I}\{t-1,t\}$ . Given subsets of the characteristics space  $Z_j\hat{I}R^K$  for  $j\hat{I}\{1,...,N\}$ , let the cartesian product  $Z=Z_1 \xrightarrow{\sim} Z_N \hat{I}R^{NK}$  be the corresponding subset of the product space. The following result concerning the conditional properties of the random variables  $X_{js}$ ,  $X_{js}$  and  $W_{js}$  holds.

*Lemma I.1*: In each period of time  $s\hat{I}$  {*t*-1,*t*}, for the set of period *s* products  $j\hat{I}$  {1,...,N} with observable characteristics  $x_{js}$  belonging to the subset of the characteristics space  $Z_j$ , the random variables  $X_{js}$ ,  $X_{js}$  and  $W_{js}$  are identically and independently distributed.

*Proof of Lemma I.1*: To prove the lemma, define the conditional distribution functions  $F_{js|Z_j} = F_{js}(x_{js})/P(x_{js} \in Z_j)$  for  $j\hat{I}\{1,...,N\}$  and  $x_{js}\hat{I}Z_j$  and the conditional distribution function  $F_{s|Z} = F_s(x_s)/P(x_s \in Z)$  for  $x_s\hat{I}Z$  (Billingsley (1986), pp. 448 ss.). Since  $x_s\hat{I}Z$  if and only if  $x_{js}\hat{I}Z_j$  for all  $j\hat{I}\{1,...,N\}$  and by the assumption of identical and independent distributions the conditioning probabilities can be factorized as:

(I.1) 
$$P(x_s \in Z) = \int_Z dF_s(x_s) = \prod_{j=1}^N \int_{Z_j} dF_{js}(x_{js}) = \prod_{j=1}^N P(x_{js} \in Z_j)$$

the following factorization of the conditional distributions holds:

(I.2) 
$$F_{s|Z}(x_s) = \prod_{j=1}^{N} F_{js|Z_j}(x_{js})$$

Similar derivations could be provided for the random variables  $X_{js}$  and  $W_j$  hence the result follows.

Lemma I.1 establishes that even after conditioning on appropriate subsets of the product space, the random variables  $X_{js}$ ,  $X_{js}$  and  $W_{js}$  remain independent and therefore conventional asymptotic theory can be applied to transformations of these random variables to study their properties. We can use this result to discuss the propositions establishing the asymptotic properties of the proposed hedonic price indexes.

*Proof of Proposition 2.1*: For each consumer, denote  $p_j$  the price that would be predicted by the model for period *t*-*1* product *j* in period *t* on the basis of the period *t* market conditions. This price includes a provision for the mark-up that would be charged for period

*t-1* product *j* in period *t*. The compensation error for each consumer is given by the difference between the compensation that would be granted on the basis of the price  $p_j$  and the compensation defined by the Laspeyres price index:

(I.3) 
$$\boldsymbol{e}_{jt-1} = (p_j - p_{jt-1}) - P_{HL}(x_{t-1}, p_{t-1}, \boldsymbol{g}_t)$$

Substituting in (I.3) the definition of the Laspeyres price index given in equation (3.2) then yields:

(I.4)  
$$\boldsymbol{e}_{jt-1} = \left( p_j - \frac{1}{N_Z} \sum_{j \in I_Z} E(c(x_{jt-1}, \boldsymbol{w}_{jt-1}, \boldsymbol{g}_t) | x_{jt-1}) \right) - \left( p_{jt-1} - \frac{1}{N_Z} \sum_{j \in I_Z} p_{jt-1} \right)$$

Averaging over products and considering that in this way the second term on the right hand side of (I.4) vanishes, we can express the average compensation error as the average of the difference between the theoretical price  $p_j$  and the price predicted on the basis of the cost function alone:

(I.5) 
$$\frac{1}{N_Z} \sum_{j \in I_Z} \boldsymbol{e}_{jt-1} = \frac{1}{N_Z} \sum_{j \in I_Z} \left( p_j - E(c(x_{jt-1}, \boldsymbol{w}_{jt-1}, \boldsymbol{g}_t) | x_{jt-1}) \right)$$

Each term in the right hand side of (I.5) can be interpreted as the compensation error for consumers of product *j* that would result by applying the compensation function (3.1) instead of the average compensation defined by the Laspeyres price index. While for each *j* this compensation error is different from  $e_{jt-1}$ , their sample averages are equal by definition.

Now note that since the theoretical price  $p_j$  is equal to the sum between predicted marginal costs  $c(x_{jt-1}, \mathbf{w}_{jt-1}, \mathbf{g})$  and the mark-up term  $m_j(x_{t-1}, \mathbf{x}_{t-1}, p, \mathbf{q}_{t-1})$ , where p is the vector of period t-1 product prices predicted by the model for period t-1 consumers given the period tcost function, each term on the right hand side of (I.5) is equal to the sum of an unobservable cost component  $\mathbf{z}_{jt-1} = c(x_{jt-1}, \mathbf{w}_{jt-1}, \mathbf{g}_t) - E(c(x_{jt-1}, \mathbf{w}_{jt-1}, \mathbf{g}_t | x_{jt-1})$  and the mark-up term  $m_j(x_{t-1}, \mathbf{x}_{t-1}, p, \mathbf{q}_{t-1})$ . It follows that the average compensation error can be expressed as the sum of the sample average of the unobserved cost component and the sample average of the mark-up term:

(I.6) 
$$\frac{1}{N_Z} \sum_{j \in I_Z} \boldsymbol{e}_{jt-1} = \frac{1}{N_Z} \sum_{j \in I_Z} \boldsymbol{z}_{jt-1} + \frac{1}{N_Z} \sum_{j \in I_Z} m_j(x_{t-1}, \boldsymbol{x}_{t-1}, p, \boldsymbol{q}_{t-1})$$

The assumption of continuity of the marginal cost function implies that the cost function is measurable, its conditional expected value by construction is also measurable and therefore  $\mathbf{z}_{jt-1}$  is a well defined random variable that satisfies usual orthogonality conditions. From assumption I.1 and lemma I.1 these regularity conditions imply that  $\mathbf{z}_{jt-1}$  is identically and independently distributed across products with mean equal to zero in each time period and therefore its sample average converges almost surely to zero as  $N_Z \rightarrow +\infty$  by the strong law of large numbers (Billingsley 1986, Theorem 22.1; White 1984, Theorem 3.1). In turn, this implies that the average compensation error converges almost surely to the average mark-up, which is greater than or equal to zero for all characteristics vectors  $(x_{t-1}, \mathbf{x}_{t-1})\hat{\mathbf{I}} R^{K+1}$  by construction:

(I.7) 
$$\frac{1}{N_Z} \sum_{j \in I_Z} \boldsymbol{e}_{jt-1} \xrightarrow{a.s.} \frac{1}{N_Z} \sum_{j \in I_Z} m_j(x_{t-1}, \boldsymbol{x}_{t-1}, p, \boldsymbol{q}_{t-1}) \ge 0$$

*Proof of Proposition 2.2:* Follows along the same lines as the proof of proposition 2.1 by suitable changes in notation.

### References

- Aizcorbe A., Corrado C. and Doms M. (2000) "Constructing Price and Quantity Indexes for High Technology Goods", Industrial Output Section, Division of Research and Statistics, Federal Reserve Board, Washington D. C. 20551.
- Anderson S. P., de Palma A. and Thisse J-F (1992) *Discrete Choice Theory of Product Differentiation*, Cambridge MA, London England, The MIT Press.
- Becker G. S. (1965) "A Theory of the Allocation of Time", *The Economic Journal*, Vol. 75, No. 299, September, pp. 497-517.
- Berndt E. R. and Griliches Z. (1993) "Price Indexes for Microcomputers: An Exploratory Study", in Foss, Manser and Young (1993), pp. 63-99.
- Berndt E. R. and Rappaport N. J. (2001) "Price and Quality of Desktop and Mobile Personal Computers: a Quarter Century Historical Overview", *American Economic Review*, Vol. 91, No. 2, May, pp. 268-273.
- Berndt E. R., Griliches Z. and Rappaport N. J. (1995) "Econometric Estimates of Price Indexes for Personal Computers in the 1990's", *Journal of Econometrics*, Vol. 68, No. 1, July, pp. 243-268.
- Berry S. T. (1994) "Estimating Discrete-Choice Models of Product Differentiation", *Rand Journal of Economics*, Vol. 25, No. 2, Summer, pp. 242-262.
- Berry S., Levinsohn J. and Pakes A. (1995) "Automobile Prices in Market Equilibrium", *Econometrica*, Vol. 63, No. 4, July, pp. 841-890.
- Berry S., Levinsohn J. and Pakes A. (2004) "Differentiated Products Demand Systems from a Combination of Micro and Macro Data: The New Car Market", *Journal of Political Economy*, Vol. 112, No. 1, February, pp. 68-105.
- Billingsley P. (1986) *Probability and Measure*, New York, Chichester, Brisbane, Toronto and Singapore, John Wiley & Sons.
- Boskin M. J., Dulberger E. R., Gordon R. J., Griliches Z. and Jorgenson D. W. (1996) "Toward a More Accurate Measure of the Cost of Living", Final Report to the Senate Finance Committee from the Advisory Commission to Study the Consumer Price Index, December 4.
- Bresnahan T. F. (1981) "Departures from Marginal-Cost Pricing in the American Automobile Industry", *Journal of Econometrics*, Vol. 17, No. 2, November, pp. 201-207.
- Bresnahan T. F. (1987) "Competition and Collusion in the American Automobile Industry: the 1955 Price War", *The Journal of Industrial Economics*, Vol. 35, No. 4, June, pp. 457-482.

- Cardell N. S. (1997) "Variance Components Structures for the Extreme-Value and Logistic Distributions with Applications to Models of Heterogeneity", *Econometric Theory*, Vol. 13, No. 2, April, pp. 185-213.
- Chow G. C. (1967) "Technological Change and the Demand for Computers", *American Economic Review*, Vol. 57, No. 5, December, pp. 1117-1130.
- Dorfman R., Samuelson P. A. and Solow R. (1958) *Linear Programming and Economic Analysis*, New York, McGraw-Hill.
- Foss F. M., Manser M. E. and Young A. H. (eds.) (1993) Price Measurement and their Uses, NBER Studies in Income and Wealth, Vol. 57, Chicago and London, University of Chicago Press.
- Goffe W. L., Ferrier G. D. and Rogers J. (1994) "Global Optimization of Statistical Functions with Simulated Annealing", *Journal of Econometrics*, Vol. 60, No.1-2, January-February, pp. 65-99.
- Gordon R. J. (1990) *The Measurement of Durable Goods Prices*, Chicago and London, University of Chicago Press.
- Gorman W. M. (1956) "A Possible Procedure for Analysing Quality Differentials in the Egg Market", published in *Review of Economic Studies*, Vol. 47, No. 150, October 1980, pp. 843-856.
- Griliches Z. (1961) "Hedonic Price Indexes for Automobiles: an Econometric Analysis of Quality Change", in *The Price Statistics of the Federal Government*, National Bureau of Economic Research, General Series, No. 73.
- Griliches Z. (1971) (ed.) Price Indexes and Quality Change: Studies in New Methods of Measurement, Cambridge, Massachusetts, Harvard University Press.
- Hansen L. P. (1982) "Large Sample Properties of Generalized Method of Moments Estimators", *Econometrica*, Vol. 50, No. 4, July, pp. 1029-1054.
- Hausman J. A. (1997a) "Valuation of New Goods Under Perfect and Imperfect Competition", in Bresnahan T. F. and Gordon R. J. (eds.) *The Economics of New Goods*, Chicago and London, The University of Chicago Press, pp. 209-247.
- Hausman J. A. (1997b) "Valuing the Effects of Regulation on New Services in Telecommunications", *Brooking Papers on Economic Activity*, Microeconomics, pp. 1-38.
- Hausman J. A. (1999) "Cellular Telephone, New Products, and the CPI", *Journal of Business and Economic Statistics*, Vol. 17, No. 2, April, pp. 188-194.
- Jorgenson D. W. and Landau R. (1989) (eds.) *Technology and Capital Formation*, Cambridge Mass., MIT Press.
- Judd K. L. (1998) Numerical Methods in Economics, The MIT Press, Cambridge MA.
- Lancaster K. J. (1966) "A New Approach to Consumer Theory", Journal of Political Economy, Vol. 74, No. 2, April, pp. 132-157.

- Lebow D. E. and Rudd J. B. (2003) "Measurement Error in the Consumer Price Index: Where Do We Stand?", *The Journal of Economic Literature*, Vol. 41, No. 1, March, pp. 159-201.
- McFadden D. (1974) "Conditional Logit Analysis of Qualitative Choice Behaviour", in Zarembka P. (ed.) *Frontiers in Econometrics*, New York and London, pp. 105-142.
- McFadden D. (1978) "Modeling the Choice of Residential Location", in Karlgvist A., et. al. (eds.) Spatial Interaction Theory and Planning Models, Amsterdam, North Holland, pp. 75-96.
- McFadden D. (1981) "Econometric Models of Probabilistic Choice" in Manski C. F. and McFadden D. (ed.) Structural Analysis of Discrete Data with Econometric Applications, The MIT Press, Cambridge MA, pp. 198-423.
- McFadden D. (1984) "Econometric Analysis of Qualitative Response Models", in Griliches Z. and Intriligator M. D. (eds.) *Handbook of Econometrics*, Vol. II, Elsevier Science Publishers BV, pp. 1395-1457.
- Moch D. (2001) "Price Indices for Information and Communication Technology Industries-An Application to the German PC Market", Discussion Paper No. 01-20, ZEW Centre For European Economic Research, March.
- Nevo A. (2003) "New Products, Quality Changes, and Welfare Measures Computed from Estimated Demand Systems", *The Review of Economics and Statistics*, May, Vol. 85, No. 2, pp. 266-275.
- Pakes A. (2003) "A Reconsideration of Hedonic Price Indices with an Application to PC's", *American Economic Review*, Vol. 93, No. 5, December, pp. 1578-1596.
- Pakes A., Berry S., Levinsohn J. A. (1993) "Applications and Limitations of Some Recent Advances in Empirical Industrial Organization: Price Indexes and the Analysis of Environmental Change", *American Economic Review*, Vol. 83, No. 2, May, pp. 240-246.
- Rosen S. (1974) "Hedonic Prices and Implicit Markets: Product Differentiation in Pure Competition", *Journal of Political Economy*, Vol. 82, No. 1, January/February, pp. 34-55.
- Triplett J. E. (1969) "Automobiles and Hedonic Quality Measurement", *Journal of Political Economy*, Vol. 77, No. 3, May/June, pp. 408-17.
- Visco I. (1994) "Caratteri Strutturali dell'Inflazione Italiana (1986-1991)", in Carlo Dell'Aringa (ed.) *Caratteri Strutturali dell'Inflazione Italiana*, il Mulino, Bologna, pp. 19-59.
- White H. (1984) Asymptotic Theory for Econometricians, Orlando, San Diego, San Francisco, New York, London, Toronto, Montreal, Sydney, Tokyo, Sao Paulo, Academic Press, Inc..
- Wynne M. A. and Rodríguez-Palenzuela D. (2004) "Measurement Bias in the HICP: What Do We Know and What Do We Need to Know?", *Journal of Economic Surveys*, Vol. 18, No. 1, February, pp.79-112.

#### RECENTLY PUBLISHED "TEMI" (\*).

- N. 522 Tecnologia e dinamica dei vantaggi comparati: un confronto fra quattro regioni italiane, by C. BENTIVOGLI and F. QUINTILIANI (September 2004).
- N. 523 The short-term impact of government budgets on prices: evidence from macroeconometric models, by J. HENRY, P. HERNÀNDEZ DE Cos and S. MOMIGLIANO, (October 2004).
- N. 524 Pricing behavior and the comovement of productivity and labor: evidence from *firm-level data*, by D.J. MARCHETTI and F. NUCCI (December 2004).
- N. 525 Is there a cost channel of monetary policy transmission? An investigation into the pricing behaviour of 2,000 firms, by E. GAIOTTI and A. SECCHI (December 2004).
- N. 526 Foreign direct investment and agglomeration: Evidence from Italy, by R. BRONZINI (December 2004).
- N. 527 Endogenous growth in open economies: A survey, by A. F. Pozzolo (December 2004).
- N. 528 The role of guarantees in bank lending, by A. F. Pozzolo (December 2004).
- N. 529 *Does the ILO definition capture all unemployment*, by A. BRANDOLINI, P. CIPOLLONE and E. VIVIANO (December 2004).
- N. 530 Household wealth distribution in Italy in the 1990s, by A. BRANDOLINI, L. CANNARI, G. D'ALESSIO and I. FAIELLA (December 2004).
- N. 531 Cyclical asymmetry in fiscal policy, debt accumulation and the Treaty of Maastricht, by F. BALASSONE and M. FRANCESE (December 2004).
- N. 532 L'introduzione dell'euro e la divergenza tra inflazione rilevata e percepita, by P. DEL GIOVANE and R. SABBATINI (December 2004).
- N. 533 A micro simulation model of demographic development and households' economic behavior in Italy, by A. ANDO and S. NICOLETTI ALTIMARI (December 2004).
- N. 534 Aggregation bias in macro models: does it matter for the euro area?, by L. MONTEFORTE (December 2004).
- N. 535 Entry decisions and adverse selection: an empirical analysis of local credit markets, by G. GOBBI and F. LOTTI (December 2004).
- N. 536 An empirical investigation of the relationship between inequality and growth, by P. PAGANO (December 2004).
- N. 537 Monetary policy impulses, local output and the transmission mechanism, by M. CARUSO (December 2004).
- N. 538 An empirical micro matching model with an application to Italy and Spain, by F. PERACCHI and E.VIVIANO (December 2004).
- N. 539 La crescita dell'economia italiana negli anni novanta tra ritardo tecnologico e rallentamento della produttività, by A. BASSANETTI, M. IOMMI, C. JONA-LASINIO and F. ZOLLINO (December 2004).
- N. 540 Cyclical sensitivity of fiscal policies based on real-time data, by L. FORNI and S. MOMIGLIANO (December 2004).
- N. 541 L'introduzione dell'euro e le politiche di prezzo: analisi di un campione di dati individuali, by E. GAIOTTI and F. LIPPI (February 2005).
- N. 542 How do banks set interest rates?, by L. GAMBACORTA (February 2005).
- N. 543 Maxmin portfolio choice, by M. TABOGA (February 2005).
- N. 544 Forecasting output growth and inflation in the euro area: are financial spreads useful?, by A. NOBILI (February 2005).
- N. 545 Can option smiles forecast changes in interest rates? An application to the US, the UK and the Euro Area, by M. PERICOLI (February 2005).
- N. 546 The role of risk aversion in predicting individual behavior, by L. Guiso and M. PAIELLA (February 2005).

<sup>(\*)</sup> Requests for copies should be sent to:

Banca d'Italia – Servizio Studi – Divisione Biblioteca e pubblicazioni – Via Nazionale, 91 – 00184 Rome (fax 0039 06 47922059). They are available on the Internet www.bancaditalia.it.

- L. GUISO and G. PARIGI, *Investment and demand uncertainty*, Quarterly Journal of Economics, Vol. 114 (1), pp. 185-228, **TD No. 289** (November 1996).
- A. F. POZZOLO, *Gli effetti della liberalizzazione valutaria sulle transazioni finanziarie dell'Italia con l'estero*, Rivista di Politica Economica, Vol. 89 (3), pp. 45-76, **TD No. 296 (February 1997)**.
- A. CUKIERMAN and F. LIPPI, Central bank independence, centralization of wage bargaining, inflation and unemployment: theory and evidence, European Economic Review, Vol. 43 (7), pp. 1395-1434, TD No. 332 (April 1998).
- P. CASELLI and R. RINALDI, *La politica fiscale nei paesi dell'Unione europea negli anni novanta*, Studi e note di economia, (1), pp. 71-109, **TD No. 334 (July 1998)**.
- A. BRANDOLINI, The distribution of personal income in post-war Italy: Source description, data quality, and the time pattern of income inequality, Giornale degli economisti e Annali di economia, Vol. 58 (2), pp. 183-239, TD No. 350 (April 1999).
- L. GUISO, A. K. KASHYAP, F. PANETTA and D. TERLIZZESE, Will a common European monetary policy have asymmetric effects?, Economic Perspectives, Federal Reserve Bank of Chicago, Vol. 23 (4), pp. 56-75, TD No. 384 (October 2000).

- P. ANGELINI, Are banks risk-averse? Timing of the operations in the interbank market, Journal of Money, Credit and Banking, Vol. 32 (1), pp. 54-73, **TD No. 266 (April 1996).**
- F. DRUDI and R: GIORDANO, *Default Risk and optimal debt management*, Journal of Banking and Finance, Vol. 24 (6), pp. 861-892, **TD No. 278 (September 1996)**.
- F. DRUDI and R. GIORDANO, *Wage indexation, employment and inflation,* Scandinavian Journal of Economics, Vol. 102 (4), pp. 645-668, **TD No. 292 (December 1996)**.
- F. DRUDI and A. PRATI, *Signaling fiscal regime sustainability*, European Economic Review, Vol. 44 (10), pp. 1897-1930, **TD No. 335 (September 1998)**.
- F. FORNARI and R. VIOLI, The probability density function of interest rates implied in the price of options, in: R. Violi, (ed.), Mercati dei derivati, controllo monetario e stabilità finanziaria, Il Mulino, Bologna, TD No. 339 (October 1998).
- D. J. MARCHETTI and G. PARIGI, Energy consumption, survey data and the prediction of industrial production in Italy, Journal of Forecasting, Vol. 19 (5), pp. 419-440, TD No. 342 (December 1998).
- A. BAFFIGI, M. PAGNINI and F. QUINTILIANI, Localismo bancario e distretti industriali: assetto dei mercati del credito e finanziamento degli investimenti, in: L.F. Signorini (ed.), Lo sviluppo locale: un'indagine della Banca d'Italia sui distretti industriali, Donzelli, TD No. 347 (March 1999).
- A. SCALIA and V. VACCA, Does market transparency matter? A case study, in: Market Liquidity: Research Findings and Selected Policy Implications, Basel, Bank for International Settlements, TD No. 359 (October 1999).
- F. SCHIVARDI, *Rigidità nel mercato del lavoro, disoccupazione e crescita*, Giornale degli economisti e Annali di economia, Vol. 59 (1), pp. 117-143, **TD No. 364 (December 1999)**.
- G. BODO, R. GOLINELLI and G. PARIGI, *Forecasting industrial production in the euro area*, Empirical Economics, Vol. 25 (4), pp. 541-561, **TD No. 370 (March 2000)**.
- F. ALTISSIMO, D. J. MARCHETTI and G. P. ONETO, *The Italian business cycle: Coincident and leading indicators and some stylized facts*, Giornale degli economisti e Annali di economia, Vol. 60 (2), pp. 147-220, **TD No. 377 (October 2000)**.

- C. MICHELACCI and P. ZAFFARONI, (*Fractional*) *Beta convergence*, Journal of Monetary Economics, Vol. 45, pp. 129-153, **TD No. 383 (October 2000)**.
- R. DE BONIS and A. FERRANDO, The Italian banking structure in the nineties: testing the multimarket contact hypothesis, Economic Notes, Vol. 29 (2), pp. 215-241, TD No. 387 (October 2000).

- M. CARUSO, Stock prices and money velocity: A multi-country analysis, Empirical Economics, Vol. 26 (4), pp. 651-72, TD No. 264 (February 1996).
- P. CIPOLLONE and D. J. MARCHETTI, *Bottlenecks and limits to growth: A multisectoral analysis of Italian industry*, Journal of Policy Modeling, Vol. 23 (6), pp. 601-620, **TD No. 314 (August 1997)**.
- P. CASELLI, Fiscal consolidations under fixed exchange rates, European Economic Review, Vol. 45 (3), pp. 425-450, TD No. 336 (October 1998).
- F. ALTISSIMO and G. L. VIOLANTE, Nonlinear VAR: Some theory and an application to US GNP and unemployment, Journal of Applied Econometrics, Vol. 16 (4), pp. 461-486, TD No. 338 (October 1998).
- F. NUCCI and A. F. POZZOLO, *Investment and the exchange rate*, European Economic Review, Vol. 45 (2), pp. 259-283, **TD No. 344 (December 1998)**.
- L. GAMBACORTA, On the institutional design of the European monetary union: Conservatism, stability pact and economic shocks, Economic Notes, Vol. 30 (1), pp. 109-143, **TD No. 356 (June 1999)**.
- P. FINALDI RUSSO and P. ROSSI, Credit costraints in italian industrial districts, Applied Economics, Vol. 33 (11), pp. 1469-1477, TD No. 360 (December 1999).
- A. CUKIERMAN and F. LIPPI, *Labor markets and monetary union: A strategic analysis*, Economic Journal, Vol. 111 (473), pp. 541-565, **TD No. 365 (February 2000)**.
- G. PARIGI and S. SIVIERO, An investment-function-based measure of capacity utilisation, potential output and utilised capacity in the Bank of Italy's quarterly model, Economic Modelling, Vol. 18 (4), pp. 525-550, TD No. 367 (February 2000).
- F. BALASSONE and D. MONACELLI, *Emu fiscal rules: Is there a gap?*, in: M. Bordignon and D. Da Empoli (eds.), Politica fiscale, flessibilità dei mercati e crescita, Milano, Franco Angeli, **TD No.** 375 (July 2000).
- A. B. ATKINSON and A. BRANDOLINI, Promise and pitfalls in the use of "secondary" data-sets: Income inequality in OECD countries, Journal of Economic Literature, Vol. 39 (3), pp. 771-799, TD No. 379 (October 2000).
- D. FOCARELLI and A. F. POZZOLO, The determinants of cross-border bank shareholdings: An analysis with bank-level data from OECD countries, Journal of Banking and Finance, Vol. 25 (12), pp. 2305-2337, TD No. 381 (October 2000).
- M. SBRACIA and A. ZAGHINI, *Expectations and information in second generation currency crises models*, Economic Modelling, Vol. 18 (2), pp. 203-222, **TD No. 391 (December 2000)**.
- F. FORNARI and A. MELE, Recovering the probability density function of asset prices using GARCH as diffusion approximations, Journal of Empirical Finance, Vol. 8 (1), pp. 83-110, TD No. 396 (February 2001).
- P. CIPOLLONE, La convergenza dei salari manifatturieri in Europa, Politica economica, Vol. 17 (1), pp. 97-125, **TD No. 398 (February 2001)**.
- E. BONACCORSI DI PATTI and G. GOBBI, *The changing structure of local credit markets: Are small businesses special?*, Journal of Banking and Finance, Vol. 25 (12), pp. 2209-2237, **TD No. 404** (June 2001).
- G. MESSINA, Decentramento fiscale e perequazione regionale. Efficienza e redistribuzione nel nuovo sistema di finanziamento delle regioni a statuto ordinario, Studi economici, Vol. 56 (73), pp. 131-148, TD No. 416 (August 2001).

- R. CESARI and F. PANETTA, Style, fees and performance of Italian equity funds, Journal of Banking and Finance, Vol. 26 (1), TD No. 325 (January 1998).
- L. GAMBACORTA, Asymmetric bank lending channels and ECB monetary policy, Economic Modelling, Vol. 20 (1), pp. 25-46, **TD No. 340 (October 1998)**.
- C. GIANNINI, "Enemy of none but a common friend of all"? An international perspective on the lenderof-last-resort function, Essay in International Finance, Vol. 214, Princeton, N. J., Princeton University Press, TD No. 341 (December 1998).
- A. ZAGHINI, Fiscal adjustments and economic performing: A comparative study, Applied Economics, Vol. 33 (5), pp. 613-624, TD No. 355 (June 1999).
- F. ALTISSIMO, S. SIVIERO and D. TERLIZZESE, *How deep are the deep parameters?*, Annales d'Economie et de Statistique,.(67/68), pp. 207-226, **TD No. 354 (June 1999)**.
- F. FORNARI, C. MONTICELLI, M. PERICOLI and M. TIVEGNA, *The impact of news on the exchange rate of the lira and long-term interest rates*, Economic Modelling, Vol. 19 (4), pp. 611-639, **TD No. 358** (October 1999).
- D. FOCARELLI, F. PANETTA and C. SALLEO, *Why do banks merge?*, Journal of Money, Credit and Banking, Vol. 34 (4), pp. 1047-1066, **TD No. 361 (December 1999)**.
- D. J. MARCHETTI, Markup and the business cycle: Evidence from Italian manufacturing branches, Open Economies Review, Vol. 13 (1), pp. 87-103, **TD No. 362 (December 1999)**.
- F. BUSETTI, *Testing for stochastic trends in series with structural breaks*, Journal of Forecasting, Vol. 21 (2), pp. 81-105, TD No. 385 (October 2000).
- F. LIPPI, *Revisiting the Case for a Populist Central Banker*, European Economic Review, Vol. 46 (3), pp. 601-612, **TD No. 386 (October 2000)**.
- F. PANETTA, The stability of the relation between the stock market and macroeconomic forces, Economic Notes, Vol. 31 (3), TD No. 393 (February 2001).
- G. GRANDE and L. VENTURA, Labor income and risky assets under market incompleteness: Evidence from Italian data, Journal of Banking and Finance, Vol. 26 (2-3), pp. 597-620, TD No. 399 (March 2001).
- A. BRANDOLINI, P. CIPOLLONE and P. SESTITO, *Earnings dispersion, low pay and household poverty in Italy, 1977-1998*, in D. Cohen, T. Piketty and G. Saint-Paul (eds.), The Economics of Rising Inequalities, pp. 225-264, Oxford, Oxford University Press, **TD No. 427** (November 2001).
- L. CANNARI and G. D'ALESSIO, La distribuzione del reddito e della ricchezza nelle regioni italiane, Rivista Economica del Mezzogiorno (Trimestrale della SVIMEZ), Vol. XVI (4), pp. 809-847, Il Mulino, TD No. 482 (June 2003).

- F. SCHIVARDI, *Reallocation and learning over the business cycle,* European Economic Review, , Vol. 47 (1), pp. 95-111, **TD No. 345 (December 1998)**.
- P. CASELLI, P. PAGANO and F. SCHIVARDI, Uncertainty and slowdown of capital accumulation in Europe, Applied Economics, Vol. 35 (1), pp. 79-89, **TD No. 372 (March 2000).**
- P. ANGELINI and N. CETORELLI, *The effect of regulatory reform on competition in the banking industry*, Federal Reserve Bank of Chicago, Journal of Money, Credit and Banking, Vol. 35, pp. 663-684, **TD No. 380 (October 2000)**.

- P. PAGANO and G. FERRAGUTO, Endogenous growth with intertemporally dependent preferences, Contribution to Macroeconomics, Vol. 3 (1), pp. 1-38, **TD No. 382 (October 2000).**
- P. PAGANO and F. SCHIVARDI, *Firm size distribution and growth*, Scandinavian Journal of Economics, Vol. 105 (2), pp. 255-274, **TD No. 394 (February 2001)**.
- M. PERICOLI and M. SBRACIA, A Primer on Financial Contagion, Journal of Economic Surveys, Vol. 17 (4), pp. 571-608, TD No. 407 (June 2001).
- M. SBRACIA and A. ZAGHINI, *The role of the banking system in the international transmission of shocks*, World Economy, Vol. 26 (5), pp. 727-754, **TD No. 409 (June 2001)**.
- E. GAIOTTI and A. GENERALE, Does monetary policy have asymmetric effects? A look at the investment decisions of Italian firms, Giornale degli Economisti e Annali di Economia, Vol. 61 (1), pp. 29-59, TD No. 429 (December 2001).
- L. GAMBACORTA, The Italian banking system and monetary policy transmission: evidence from bank level data, in: I. Angeloni, A. Kashyap and B. Mojon (eds.), Monetary Policy Transmission in the Euro Area, Cambridge, Cambridge University Press, TD No. 430 (December 2001).
- M. EHRMANN, L. GAMBACORTA, J. MARTÍNEZ PAGÉS, P. SEVESTRE and A. WORMS, *Financial systems and the role of banks in monetary policy transmission in the euro area*, in: I. Angeloni, A. Kashyap and B. Mojon (eds.), Monetary Policy Transmission in the Euro Area, Cambridge, Cambridge University Press, **TD No. 432 (December 2001)**.
- F. SPADAFORA, Financial crises, moral hazard and the speciality of the international market: further evidence from the pricing of syndicated bank loans to emerging markets, Emerging Markets Review, Vol. 4 (2), pp. 167-198, TD No. 438 (March 2002).
- D. FOCARELLI and F. PANETTA, Are mergers beneficial to consumers? Evidence from the market for bank deposits, American Economic Review, Vol. 93 (4), pp. 1152-1172, **TD No. 448 (July 2002)**.
- E.VIVIANO, Un'analisi critica delle definizioni di disoccupazione e partecipazione in Italia, Politica Economica, Vol. 19 (1), pp. 161-190, **TD No. 450** (July 2002).
- M. PAGNINI, Misura e Determinanti dell'Agglomerazione Spaziale nei Comparti Industriali in Italia, Rivista di Politica Economica, Vol. 3 (4), pp. 149-196, **TD No. 452 (October 2002)**.
- F. BUSETTI and A. M. ROBERT TAYLOR, Testing against stochastic trend and seasonality in the presence of unattended breaks and unit roots, Journal of Econometrics, Vol. 117 (1), pp. 21-53, TD No. 470 (February 2003).

- F. LIPPI, Strategic monetary policy with non-atomistic wage-setters, Review of Economic Studies, Vol. 70 (4), pp. 909-919, TD No. 374 (June 2000).
- P. CHIADES and L. GAMBACORTA, *The Bernanke and Blinder model in an open economy: The Italian case*, German Economic Review, Vol. 5 (1), pp. 1-34, **TD No. 388 (December 2000)**.
- M. BUGAMELLI and P. PAGANO, *Barriers to Investment in ICT*, Applied Economics, Vol. 36 (20), pp. 2275-2286, **TD No. 420 (October 2001)**.
- A. BAFFIGI, R. GOLINELLI and G. PARIGI, *Bridge models to forecast the euro area GDP*, International Journal of Forecasting, Vol. 20 (3), pp. 447-460, **TD No. 456 (December 2002)**.
- D. AMEL, C. BARNES, F. PANETTA and C. SALLEO, Consolidation and Efficiency in the Financial Sector: A Review of the International Evidence, Journal of Banking and Finance, Vol. 28 (10), pp. 2493-2519, TD No. 464 (December 2002).
- M. PAIELLA, *Heterogeneity in financial market participation: appraising its implications for the C-CAPM*, Review of Finance, Vol. 8, pp. 1-36, **TD No. 473 (June 2003)**.
- E. BARUCCI, C. IMPENNA and R. RENÒ, *Monetary integration, markets and regulation*, Research in Banking and Finance, (4), pp. 319-360, **TD No. 475** (June 2003).

- E. BONACCORSI DI PATTI and G. DELL'ARICCIA, *Bank competition and firm creation*, Journal of Money Credit and Banking, Vol. 36 (2), pp. 225-251, **TD No. 481 (June 2003)**.
- R. GOLINELLI and G. PARIGI, Consumer sentiment and economic activity: a cross country comparison, Journal of Business Cycle Measurement and Analysis, Vol. 1 (2), pp. 147-172, TD No. 484 (September 2003).
- L. GAMBACORTA and P. E. MISTRULLI, *Does bank capital affect lending behavior?*, Journal of Financial Intermediation, Vol. 13 (4), pp. 436-457, **TD No. 486 (September 2003)**.
- F. SPADAFORA, *Il pilastro privato del sistema previdenziale: il caso del Regno Unito*, Rivista Economia Pubblica, (5), pp. 75-114, **TD No. 503** (June 2004).
- G. GOBBI e F. LOTTI, Entry decisions and adverse selection: an empirical analysis of local credit markets, Journal of Financial services Research, Vol. 26 (3), pp. 225-244, TD No. 535 (December 2004).
- F. CINGANO e F. SCHIVARDI, *Identifying the sources of local productivity growth*, Journal of the European Economic Association, Vol. 2 (4), pp. 720-742, **TD No. 474 (June 2003)**.
- C. BENTIVOGLI e F. QUINTILIANI, Tecnologia e dinamica dei vantaggi comparati: un confronto fra quattro regioni italiane, in C. Conigliani (a cura di), Tra sviluppo e stagnazione: l'economia dell'Emilia-Romagna, Bologna, Il Mulino, **TD No. 522** (October 2004).

- A. DI CESARE, Estimating Expectations of Shocks Using Option Prices, The ICFAI Journal of Derivatives Markets, Vol. II (1), pp. 42-53, TD No. 506 (July 2004).
- M. OMICCIOLI, Il credito commerciale: problemi e teorie, in L. Cannari, S. Chiri e M. Omiccioli (a cura di), Imprese o intermediari? Aspetti finanziari e commerciali del credito tra imprese in Italia, Bologna, Il Mulino, TD No. 494 (June 2004).
- L. CANNARI, S. CHIRI e M. OMICCIOLI, Condizioni del credito commerciale e differenzizione della clientela, in L. Cannari, S. Chiri e M. Omiccioli (a cura di), Imprese o intermediari? Aspetti finanziari e commerciali del credito tra imprese in Italia, Bologna, Il Mulino, TD No. 495 (June 2004).
- P. FINALDI RUSSO e L. LEVA, Il debito commerciale in Italia: quanto contano le motivazioni finanziarie?, in L. Cannari, S. Chiri e M. Omiccioli (a cura di), Imprese o intermediari? Aspetti finanziari e commerciali del credito tra imprese in Italia, Bologna, Il Mulino, TD No. 496 (June 2004).
- A. CARMIGNANI, Funzionamento della giustizia civile e struttura finanziaria delle imprese: il ruolo del credito commerciale, in L. Cannari, S. Chiri e M. Omiccioli (a cura di), Imprese o intermediari? Aspetti finanziari e commerciali del credito tra imprese in Italia, Bologna, Il Mulino, TD No. 497 (June 2004).
- G. DE BLASIO, Does trade credit substitute for bank credit?, in L. Cannari, S. Chiri e M. Omiccioli (a cura di), Imprese o intermediari? Aspetti finanziari e commerciali del credito tra imprese in Italia, Bologna, Il Mulino, TD No. 498 (June 2004).
- M. BENVENUTI e M. GALLO, Perché le imprese ricorrono al factoring? Il caso dell'Italia, in L. Cannari, S. Chiri e M. Omiccioli (a cura di), Imprese o intermediari? Aspetti finanziari e commerciali del credito tra imprese in Italia, Bologna, Il Mulino, TD No. 518 (October 2004).